Author's response to reviews

Title: A multidisciplinary clinical treatment of locally advanced rectal cancer complicated by rectovesical fistula: a case report.

Authors:

Tiancheng Zhan (jessecrash@sina.com.cn)
Lin Wang (wanglinmd@gmail.com)
Ming Li (limingmd@126.com)
Zhongwu Li (zhwuli2008@sina.com)
Yong Cai (caiyong109@souhu.com)
Lin Shen (lin100@medmail.com.cn)
Jin Gu (zlguj@yahoo.com.cn)

Version: 6 Date: 13 September 2012

Author's response to reviews: see over
Journal of Medical Case Reports Editorial

MS: 1663666366994498
A multidisciplinary clinical treatment of locally advanced rectal cancer complicated with rectovesical fistula: a case report.
Tiancheng Zhan, Lin Wang, Ming Li, Zhongwu Li, Yong Cai, Lin Shen and Jin Gu

Dear Editor,

I am sending you a revised manuscript, which was checked and edited by a native English speaker according your suggestions. We believe that the quality of language of the manuscript is of high standard now.

We also addressed those comments raised by the reviews in our revised manuscript. We added the option of performing cystoscopy for the diagnosis of residual tumor in the bladder in the discussion section (highlighted in red) according to reviewer 1 and corrected the grammatical errors in the manuscript, specifically re-wrote the sentence in the third paragraph under the heading of “case presentation” (highlighted in red) according to review 2.

Thank you very much for your kind considering publication of this manuscript in The Journal of Medical Case Reports.

Best regards,

Jin Gu

Your title